Author's response to reviews

Title: Pragmatic randomized controlled trial of individually prescribed exercise versus usual care in a heterogeneous cancer survivor population. A feasibility study

Authors:

Julie M Walsh (walshj7@tcd.ie)
Juliette Hussey (jmhussey@tcd.ie)
Emer Guinan (emguinan@tcd.ie)
Dearbhaile O’ Donnell (dodonnell@stjames.ie)

Version: 2  Date: 23 January 2010

Author's response to reviews:

Dear Dr. Simmons

Please find that suggested revisions which have been completed - these are both under the ´Patient recruitment´ section paragraph under the "Methods/Design” section

1. A statement in the manuscript stating that informed consent from each subjects will be obtained is included

2. The name of the review board that granted ethical approval is also included

I hope all is in order, however should there be any further amendments you wish to be included, please do not hesitate to let me know,

Best Wishes, Julie